Author's response to reviews

Title: Lemon juice has protective activity in a rat urolithiasis model

Authors:

Mohammed Touhami (mohammed.touhami@gmail.com)
Amine Laroubi (lar_amine@hotmail.com)
Khadija Elhabazi (kelhabazi@hotmail.com)
Farouk Loubna (loubna_fa@hotmail.com)
Ibtissam Zrara (ibtissam.zrara@yahoo.fr)
Younes Eljahiri (younes.eljahiri@yahoo.fr)
Abdelkhalek Oussama (oussamaabdelkhalek@yahoo.fr)
Felix Grases (fgrases@uib.es)
Abderrahman Chait (chait@ucam.ac.ma)

Version: 4 Date: 24 September 2007

Author's response to reviews: see over
24th September 2007

Dear Editor,

Thank you very much for accepting the paper entitled “Lemon juice has protective activity in a rat urolithiasis model”. I am sending you the version of the manuscript including the requested minor changes.

Being at your disposal, yours faithfully,

Prof. Dr. F. Grases